\DocumentMetadata{
	lang=en,
	pdfstandard=ua-2,
	tagging=on,
	tagging-setup={math/setup=mathml-SE}
	}

\documentclass [12pt] {article}


\usepackage{graphicx, amsmath}
\usepackage{enumext}

\DeclareGraphicsRule{.tif}{png}{.png}{`convert #1 `dirname #1`/`basename #1 .tif`.png}


\topmargin 0pt
\advance \topmargin by -\headheight
\advance \topmargin by -\headsep
\textheight 8.9in
\oddsidemargin 0pt
\evensidemargin \oddsidemargin
\marginparwidth 0.5in
\textwidth 6.5in

\title{Quiz 0}
\date{}

\begin{document}

\maketitle%\thispagestyle{plain}


\vspace{-0.9in}
\noindent This is a two-stage quiz. During the first stage, use your knowledge \& calculator to take this quiz. You have 15 min. In the second stage, you are now welcome to use your books, notes, and students in the class to retake the same quiz. You have 15 min. to write one solution (with everyone's name on it!) to be turned in for the group. 


\vspace{.3in}

\noindent{Show \emph{all} your work. Reasonable supporting work must be shown for any partial credit. }

\begin{enumext}
\item $[2]$ What career(s) are you \emph{interested} in at this point? \\
\vspace{.6 in}

\item $[2]$ Write down something that the person (if stage 1)/people (if stage 2) taking this quiz are good at. It does not have to be academic.\\
\vspace{1.25in}

\item $[3]$ What advice would you share to someone who wants to get good at the same thing you are good at?\\
\vspace{1.25in}

\item $[3]$  What do you want the instructor to know about you? \newline If this is stage 2 of the quiz, what do you want other students to know about you? (There could be one thing shared by everyone or 3 separate items.)
\vspace{1in}


\vspace{1.0in}

\end{enumext}
\end{document}
